Author’s response to reviews

Title: Proteinuria as a presenting sign of combined methylmalonic acidemia and homocysteinemia: case report

Authors:

Ruyue CHEN (781135340@qq.com)
Xiao-zhong LI (xiaozhonglicn@yeah.net)
Qiang LIN (wslinqiang@aliyun.com)
Yun ZHU (ruyuechen@foxmail.com)
Yun-yan SHEN (syunyan@sohu.com)
Qin-ying XU (listenway_li@163.com)
Xue-ming ZHU (xueming_zhu@aliyun.com)

Version: 1 Date: 22 Oct 2019

Author’s response to reviews:

Ethics Approval and Consent to Participate
Verbal informed consent was obtained from all individual participants included in the study. All procedures performed in studies involving human participants were in accordance with the ethical standards of the Medical ethics committee in Children’s Hospital of Soochow University. The study was submitted to and approved by the committee.

Consent for publication
Written informed consent for publication of clinical details and/or clinical images was obtained from all of the participants and from the parents or legal guardians of the minors in case.